Author's response to reviews

Title: An electronic clinical trial tool to recruit large patient samples and assess selection bias in general practice research

Authors:

Stephanie Heinemann (sheinem3@gwdg.de)
Sabine Thüring (sthueri@gwdg.de)
Sven Wedeken (swedeken@gwdg.de)
Tobias Schäfer (tobias.schaefer@medizin.uni-goettingen.de)
Christa Scheidt-Nave (Scheidt-Nave@rki.de)
Mirko Ketterer (mirko.ketterer@it-choice.de)
Wolfgang Himmel (whimmel@gwdg.de)

Version: 5 Date: 24 January 2011

Author's response to reviews:

Dear Editorial Team,

Thank you for accepting our paper. We have only made very minor revisions.

Minor revisions:
1. On page 16 is now a detailed list of authors' contributions.
2. On page 16, we now list the affiliation of Mirko Ketterer with an IT company under Conflicts of Interest.
3. Table 2 was re-formatted because the decimal places were separated with a comma instead of a full stop. (This is a typical transformation error between German and English. Sorry we missed it in the previous version.)
4. On page 22, we give a legend for Figure 1, which explains that "Hans Mustermann" is an alias. By the way, this is the German form of "John Doe".

Best regards,

Stephanie Heinemann and Wolfgang Himmel (for the authors)